Author's response to reviews

Title: Delay in diagnosis of muscle disorders depends on the subspecialty of the initially consulted physician

Authors:

Simone Spuler (simone.spuler@charite.de)
Andrea Stroux (andrea.stroux@charite.de)
Franziska Kuschel (franziska.kuschel@charite.de)
Adelheid Kuhlme (adelheid.kuhlme@charite.de)
Friederike Kendel (friederike.kendel@charite.de)

Version: 3 Date: 11 February 2011

Author's response to reviews:

Figures: The first figure referenced in the manuscript text should be Figure 1. The second, figure 2, etc. Please reorder the figure citation so that they are cited in consecutive order. All figures must be cited; figures not cited will be excluded from the final manuscript. Please do not use ranges when listing figures. Please list all the figures, for example: Figures 5, 6, 7, 8 and 9.

--> Figure 2 is now mentioned in the manuscript (p.6, line 8 from the bottom)

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

--> done